Author's response to reviews

Title: Payments and quality of care in private for-profit and public hospitals in Greece

Authors:

Elias Kondilis (eliaskon@med.auth.gr)
Magda Gavana (gavana@med.auth.gr)
Stathis Giannakopoulos (gianstat@med.auth.gr)
Emmanouil Smyrnakis (smyrnak@auth.gr)
Nikolaos Dombros (dombros@auth.gr)
Alexis Benos (benos@med.auth.gr)

Version: 3 Date: 3 June 2011

Author's response to reviews:

Dear sirs

we submit a revised version of our research paper under the title "Payments and quality of care in private for-profit and public hospitals in Greece". The changes made in our manuscript address the Associate Editor's concerns:
1. the manuscript was copyedited, with the assistance of a professional copyediting service (English Manager Science Editing)
2. the introduction of our paper has been shortened by approximately 160 words
3. double spacing has been used at the final format of our manuscript.

Sincerely yours
Dr. Elias Kondilis